Author's response to reviews

Title: CCl4 induced genotoxicity and DNA oxidative damages in rats: Hepatoprotective effect of Sonchus arvensis

Authors:

Huda Alkreathy (halkreathy@gmail.com)
rahmat A Khan (rahmatgul_81@yahoo.com)
Muhammad Rashid Khan (mrkhangau@yahoo.com)
sumaira Sahreen (sumairasahreen@gmail.com)

Version: 2 Date: 20 January 2014

Author's response to reviews: see over
Dear Editor,

Dear Sir,

Subject: - CCl₄ induced genotoxicity and DNA oxidative damages in rats: Hepatoprotective effect of Sonchus arvensis

Dear Sir,

Reference to the Comments on Evaluation of CCl₄ induced genotoxicity and DNA oxidative damages in rats: Hepatoprotective effect of Sonchus arvensis, Enclosed, please find a revised version of our Manuscript. All points were considered in detail, and correction was carried out with red color. We would like to thank the Editors for their careful reading of the manuscript. Their comments improved the quality of the manuscript, and we hope that it will be now suitable for publication in your esteemed Journal.

Thank you for your attention.

Dr. Rahmat Ali Khan
Response to Editor Remarks

1. Please include e-mail addresses of all author's in the title page.
   
   Ans: Added

2) Please provide acknowledgment section after author's contribution.
   
   Ans: Added